Author's response to reviews

Title: New place of living, eroding resilience and psychological distress among Somali Refugees: A qualitative and quantitative study

Authors:

Nasir Warfa (n.warfa@qmul.ac.uk)
Kam Bhui (k.s.bhui@qmul.ac.uk)
Charles Watters (c.watters@rutgers.edu)
Ken Carswell (k.carswell@qmul.ac.uk)
David Ingleby (J.D.Ingleby@uu.nl)
Sarah Curtis (S.E.Curtis@durham.ac.uk)

Version: 2 Date: 19 December 2011

Author's response to reviews:

Dear Editor

We have made all the requested changes to the manuscript. We inserted in the paper, (1) a title page, (2) new section on ethics approval, (3) statement on conflict of interest and (4) information about authors' contributions.